Author's response to reviews

Title: Factors related to treatment intensity in Swiss primary care

Authors:

André Busato (andre.busato@memcenter.unibe.ch)
Pius Matter (pius.matter@vwi.unibe.ch)
Beat Künzi (beat.kuenzi@swisspep.ch)

Version: 11 Date: 13 March 2009

Author's response to reviews:

Dear Sirs

I rechecked our manuscript and I found to two minor issues in the section
Strengths and Limitations

The fist sentence was changed into

“The availability of data from a large majority of Swiss primary care providers
who are still under mandatory contracts with health insurers was a strength of the
study.”

The second sentence, second paragraph was chanced into

“As stated above the volume of consultations is likely to be higher since the first
300 CHF (#200 €) or more of treatment cost within a year has to be paid out of
pocket.”

In order to be compatible with a similar statement made in the first paragraph of
the discussion

I uploaded this version to BMC Health Services Research and I confirm that the
manuscript is now acceptable for publication.

With best regards

A. Busato